Author’s response to reviews

Title: What is the evidence base to guide surgical treatment of infected hip prostheses? Systematic review of longitudinal studies in unselected patients

Authors:

Andrew D Beswick (andy.beswick@bristol.ac.uk)
Karen T Elvers (Karen.Elvers@bristol.ac.uk)
Rachael Gooberman-Hill (R.Gooberman-Hill@bristol.ac.uk)
Alison Smith (alison.smith@bristol.ac.uk)
Andrew Lovering (Andrew.Lovering@nbt.nhs.uk)
Ashley W Blom (ashley.blom@bristol.ac.uk)

Version: 4 Date: 6 January 2012

Author’s response to reviews:

Dear sir

Submission previously with BMC Musculoskeletal Disorders

Yours faithfully

Andrew Beswick